Author's response to reviews

Title: What is Traditional Acupuncture? Exploring goals and processes of treatment in the Context of Women with Early Breast Cancer

Authors:

Sarah Price Dr (sarah.price@talk21.com)
Andrew F Long Prof (A.F.Long@leeds.ac.uk)
Mary Godfrey Dr (M.Godfrey@leeds.ac.uk)

Version: 2  Date: 10 January 2014

Author's response to reviews:

I can confirm that included in the manuscript is the LREC (Leeds (East) Research Committee) and that written informed consent was obtained from all participants and that all data has been anonymised.